Author's response to reviews

Title: The Effect of Ventricular Assist Devices on Cerebral Autoregulation: a preliminary Study

Authors:

Judith Bellapart (30489jbr@comb.es)
Gregory Chan (gregchan@unsw.edu.au)
Yu-Chieh Tzeng (shieak_tzeng@otago.ac.nz)
Philip Ainsley (Philip.inslie@ubc.ca)
Adrian G Barnett (a.barnett@qut.edu.au)
Kimble R Dunster (k.dunster@qut.edu.au)
Rob Boots (r.boots@mailbox.eq.edu.au)
John F Fraser (j.fraser@uq.edu.au)

Version: 4 Date: 18 February 2011

Author's response to reviews: see over
17th\textsuperscript{nd} February 2011

To the Editor-in-chief of BMC Anesthesiology Journal

RE: The Effect of Ventricular Assist Devices on Cerebral Autoregulation: a Preliminary Study

We write in response to the recommended modifications received on the 16\textsuperscript{th} February 2011 in regards of the editorial and formatting changes suggested for this manuscript.

Please find attached the modified manuscript (with tracked changes as requested).

Yours Sincerely

Dr Judith Bellapart

MD, ESICM, JFICM

Intensive care Specialist

Royal Brisbane Women Hospital

Herston, Brisbane, QLD 4029, Australia